Dear Editor,

Thank you for the reviewers' constructive comments. We have addressed all the comments raised by reviewers accordingly with all the changes throughout the revised manuscript (NO: 70006). Your favorable consideration is greatly appreciated.

1. References should be rewritten as per journal requirement.
   Re: Relevant references have been rewritten in the manuscript.

2. Editorial corrections had been made in the manuscript (Red strikethrough – to be deleted, Blue – to be added)
   Re: Relevant information has been deleted in the manuscript.

3. Uniform presentation should be used for figures showing the same or similar contents; for example, “Figure 1 Pathological changes of atrophic gastritis after treatment. A: ...; B: ...; C: ...; D: ...; E: ...; F: ...; G: ...”.
   Re: We provide the original figure documents using PowerPoint.